Author's response to reviews

Title: Effect of posture-control insoles on function in children with cerebral palsy: Randomized controlled clinical trial

Authors:

Hugo Pasini Neto (hugopneto@yahoo.com.br)
Luanda André Collange Grecco (luandacollange@hotmail.com)
Thaluana Cristovão (thaluannaclc@hotmail.com)
Luiz Alfredo Braun (luiz_braun@hotmail.com)
Lilian Chrystiane Giannasi (odontogiannasi@uol.com.br)
Afonso Shiguemi Inoue Salgado (afonsoisalgado@yahoo.com.br)
João CF Corrêa (jcorrea@uninove.br)
Luciana MM Sampaio (lucianamalosa@uninove.br)
Manuela Galli (manuela.galli@polimi.it)
Claudia Santos Oliveira (csantos@uninove.br)

Version: 2 Date: 11 July 2012

Author's response to reviews:

Dear Ms. Abigail Quiniquini
Journal Editorial Office

The registration information and financial support for research are described in the submitted manuscript (information described below). The number included in the abstract. Ethical and external funding approval (is also attached in the submitted manuscript).

Anyway we are sending the manuscript with registration information, the ethics committee and financial support (copies of requested documents and translation of these).

Take this opportunity to clarify that the protocol has not yet submitted results, do not submit articles concerning this protocol.

Thank you for the opportunity, we are available for further information.